SPECIFIC COMMENTS TO AUTHORS

Escitalopram, a new drug, has a rapid effect and is widely used in the treatment of depression. External treatment offers the characteristics of convenience, economy, and fewer adverse reactions and is very apt for treating depression. In this study, the authors evaluated the clinical benefits of Guipi decoction combined with escieptidine oxalate tablets for individuals with depression. The manuscript is overall well designed and the results are interesting. The methods are described in detail. Minor comments: 1. The manuscript have to be edited. Some minor language polishing should be corrected. 2. The limit of the study should be included and discussed. 3. The conclusion is missing in the main text. Please check and add it. 4. The references list should be updated.

Reviewer #1:
Specific Comments to Authors: This study is very interesting. The authors investigated the therapeutic effect of Guipi decoction plus reduced Du vein fumigation in combination with escitalopram oxalate tablets. The findings are interesting and well discussed. After a minor revision, it can be accepted.

I am glad that you have reviewed my manuscript. I will make minor modifications to the manuscript based on your feedback

Reviewer #2:
Specific Comments to Authors: Escitalopram, a new drug, has a rapid effect and is widely used in the treatment of depression. External treatment offers the characteristics of convenience, economy, and fewer adverse reactions and is very apt for treating depression. In this study, the authors evaluated the clinical benefits of Guipi decoction combined with escieptidine oxalate tablets for individuals with depression. The
manuscript is overall well designed and the results are interesting. The methods are described in detail. Minor comments: 1. The manuscript have to be edited. Some minor language polishing should be corrected. 2. The limit of the study should be included and discussed. 3. The conclusion is missing in the main text. Please check and add it. 4. The references list should be updated.

1. I have corrected some minor language modifications in the manuscript.
2. Thank you very much for your feedback. I have added the limitations of the discussion research.
3. I have checked and added the missing conclusions in the main text. The list of references has also been updated.